Author’s response to reviews

Title: Magnitude and factors associated with Intimate Partner Violence in Mainland Tanzania

Authors:
Method Kazaura (mrkazaura@muhas.ac.tz)
Mangi Ezekiel (e_mangi@yahoo.com)
Dereck Chitama (dereck.chitama@gmail.com)

Version: 3 Date: 04 May 2016

Author’s response to reviews:

4th May, 2016

The Editor,

BMC Public Health

Dear Pafitis,

We received the editorial request to indicate:

Whether the data we used for your study is openly available or whether we received permission (and from who) to use it.
Response:

Although authors in collaboration with USAID (funding and other logistical support) designed the study, executed the protocol and analyzed data and wrote the manuscript, we were authorized (a letter is available) to disseminate research findings through publication by USAID officials. In the manuscript (in the Declaration section), we have indicated that data are available by requesting USAID (Tanzania) through the authors.

Sincerely,

Method Kazaura

(On behalf of all authors)